Author's response to reviews

Title: Evaluation of subcortical grey matter abnormalities in patients with MRI-negative cortical epilepsy determined through structural and tensor magnetic resonance imaging

Authors:

Syu-Jyun Peng (blue.year@msa.hinet.net)
Tomor Harnod (nsha@tzuchi.com.tw)
Jang-Zern Tsai (jztsai@ee.ncu.edu.tw)
Ming-Dou Ker (mdker@mail.nctu.edu.tw)
Jun-Chern Chiou (chiou@mail.nctu.edu.tw)
Herming Chiueh (chiueh@mail.nctu.edu.tw)
Chung-Yu Wu (peterwu@mail.nctu.edu.tw)
Yue-Loong Hsin (hsin.yloong@msa.hinet.net)

Version: 6 Date: 9 April 2014

Author's response to reviews: see over
Dear Dr. Aidan Neligan:

We have great appreciation for you and the reviewers for the careful critiques of the manuscript (MS: 1443335159117423 - Evaluation of subcortical grey matter abnormalities in patients with MRI-negative cortical epilepsy determined through structural and tensor magnetic resonance imaging).

The term “informed consent” was modified to “written informed consent”.

Sincerely,

Yue-Loong Hsin, MD
Director, Department of Neurology Chung Shan Medical University Hospital, No.110, Sec. 1, Jianguo N. Rd., South Dist., Taichung City 40201, Taiwan
E-mail: hsin.yloong@msa.hinet.net
Tel: +886 4 24739595